Author's response to reviews

Title: Acute abdominal pain in a patient on automated peritoneal dialysis - acute appendicitis: a case report

Authors:

Robert Ekart (robert.ekart2@guest.arnes.si)
Matjaž Horvat (maco.horvat@gmail.com)
Miran Koželj (miran.kozelj@ukc-mb.si)
Breda Pečovnik Balon (breda.balon@ukc-mb.si)
Sebastjan Bevc (sebastjanb@yahoo.com)
Radovan Hojs (radovan.hojs@ukc-mb.si)

Version: 3 Date: 11 July 2012

Author's response to reviews:

Below please find our response to the reviewer`s comments:

1. The manuscript was checked and repaired by native speaker of English.
2. All changes made in manuscript are yellow coloured.
3. We have added a new sentence about the importance of a negative CT scan.
4. We have added all details about myocardial ischemia-ECG at admission, clinical picture and also enzymatic curve for troponin T was made (Figure 1).
5. The sentences regarding the gastroenteritis of the patient`s girlfriend were removed from the manuscript.